Author's response to reviews

Title: Contact Lens Rehabilitation following Repaired Corneal Perforations

Authors:

Jeewan S Titiyal (titiyal@rediffmail.com)
Rajesh Sinha (sinharaj1@rediffmail.com)
Namrata Sharma (namrata103@hotmail.com)
Rasik B Vajpayee (rasikvajpayee@rediffmail.com)
V Sreenivas (sreenivas_vishnu@yahoo.com)

Version: 2 Date: 23 December 2005

Author's response to reviews: see over
Editor

BMC Ophthalmology

RE: Submission of the manuscript, entitled “Contact Lens Rehabilitation following Repaired Corneal Perforations”

Dear Editor,

Enclosed please find the revised version of the above mentioned manuscript for consideration for publication in your journal.

I hope the paper now fulfils the requirements for publication in your esteemed journal.

With warm regards

Yours sincerely,

Rajesh Sinha
Response to reviewers’ comments

Reviewer 1:

1. As suggested by the reviewer, the typographical error regarding mean values presented for BCVA in the Abstract (Page 2, Para 3, Lines 6-7), and in the results section (Page 6, last line) have been corrected and have been reduced to two decimals.
2. Page 2, Abstract: As suggested by the reviewer we have now used the term “irregular cornea” (Page 2, Last line).
3. As suggested by the reviewer, we have specified clearly regarding the time of comparison of the visual function (Page 7, first line).
4. Page 6, Statistical analysis: As suggested by the reviewer, paired t-test has now been used to compare the visual outcome with contact lens and glass in all the patients.
5. Page 7, Results: As suggested by the reviewer, the evaluation of the differences in BCVA has now been included.
6. Page 12, Figure 1: As suggested by the reviewer, the figure has been removed.
7. Page 13, Table 1: As suggested by the reviewer, the analysis of the differences in BCVA has been re-evaluated.

Reviewer 2:

1. As suggested by the reviewer, we have now used one-way ANOVA for comparison of these groups.
2. As suggested by the reviewer, paired t-test has now been used for comparing the outcome of spectacle vs. contact lens.